Author's response to reviews

**Title:** Cognition, behaviour and academic skills after cognitive rehabilitation in Ugandan children surviving severe malaria: a randomised trial

**Authors:**

- Paul Bangirana (pbangirana@yahoo.com)
- Peter Allebeck (peter.allebeck@ki.se)
- Michael J Boivin (michael.boivin@hc.msu.edu)
- Chandy C John (ccj@umn.edu)
- Connie Page (pagec@msu.edu)
- Anna Ehnvall (anna.ehnvall@lthalland.se)
- Seggane Musisi (segganemusisi@yahoo.ca)

**Version:** 6 **Date:** 4 August 2011

**Author's response to reviews:**

1. Changed the spelling of Kaufmann to the correct Kaufman.
2. The age range in the abstract methods is 5 to 12 not 5 to 10, this has been changed.